Author's response to reviews

Title: Obesity, metabolic syndrome, impaired fasting glucose and microvascular dysfunction: a principal component analysis approach

Authors:

Diogo G Panazzolo (guarnipana@yahoo.com.br)
Fernando L Sicuro (fsicuro@gmail.com)
Ruth Clapauch (rclapauch@uol.com.br)
Priscila A Maranhão (priscilamaranhao@gmail.com)
Eliete Bouskela (eliete_bouskela@yahoo.com.br)
Luiz G Kraemer-Aguiar (gkraemer@iq.com.br)

Version: 5 Date: 17 October 2012

Author’s response to reviews:

Dear Prof. Keiran Mather

According to your suggestion and agreeing with you, we decided to exclude all figures from the manuscript. The main results of our manuscript are depicted in table 2 and the reader would be able to catch all information in it.

We have also submitted the manuscript to the Editor´s suggested site (www.edanzediting.com/bmc1) to be edited by an english native-speaker reviewer. We hope that the manuscript now meets your approval. Finally, we should express our sincere feelings of gratitude since your comments were all taken into consideration and resulted in many discussions by our group which improved our knowledge.

The final version is on “Clear Format” and the copy edited version is on “Tracked Format”.

Sincerely yours,

L. G. Kraemer-Aguiar